Author's response to reviews

Title: Eating for 1, Healthy and Active for 2; feasibility of delivering novel, compact training for midwives to build knowledge and confidence in giving nutrition, physical activity and weight management advice during pregnancy.

Authors:

   Andrea J Basu (partridgeandrea@hotmail.com)
   Lynne Kennedy Prof (l.kennedy@chester.ac.uk)
   Karen Tocque Prof (k.tocque@chester.ac.uk)
   Sharn Jones (sharn.jones@wales.nhs.uk)

Version: 4 Date: 4 June 2014

Author's response to reviews: see over
Dear Editorial Team

Re: resubmission of the following research paper to BMC Pregnancy & Childbirth:

Eating for 1, Healthy and Active for 2; feasibility of delivering novel, compact training for midwives to build knowledge and confidence in giving nutrition, physical activity and weight management advice during pregnancy.

(MS: 1085200182111018)

The authors wish to thank the editorial team for their constructive feedback and review of the manuscript. In response to the editorial comment;

"Before publication ethical considerations should be added under the heading ethic. These considerations should be based on ethical principles related to the participants involvement in the study (see reviewer 3), not on being overweight as a woman."

The authors have inserted a new paragraph within the Ethics section which we hope will address the comment raised above.

We look forward to hearing the outcome of your decision in due course,

Yours Sincerely

Andrea Basu MSc BSc (Hons) RD RNutr (Public Health).